Author's response to reviews

Title: The effects of a three-week use of lumbosacral orthoses on trunk muscle activity and their response to trunk perturbations.

Authors:

Jacek Cholewicki (cholewic@msu.edu)
Kevin C McGill (kevincraigmccgill@yahoo.com)
Krupal R Shah (krupal@aya.yale.edu)
Angela S Lee (leang@msu.edu)

Version: 5 Date: 29 June 2010

Author's response to reviews: see over
Editorial Production Team  
BMC Musculoskeletal Disorders  

June 28, 2010

RE: The effects of a three-week use of lumbosacral orthoses on trunk muscle activity and their response to trunk perturbations.

Dear Editors,

Enclosed please find our revised manuscript with following changes:

Major revisions
1. References are now cited in numerical order.
Minor revisions
2. We removed the authors' qualifications from the manuscript and submissions system.
3. Table titles are now only one sentence long. We placed any other text as a legend below the table.
4. Figures are cropped tightly and sized to fit either one or two-column width.
5. “Figure captions” section was renamed to “Figure legends”

Sincerely yours,

Jacek Cholewicki, Ph.D.  
Walter F. Patenge Professor and  
Ingham Foundation Research Fellow  
Dept. of Surgical Specialties  
College of Osteopathic Medicine  
Michigan State University

Address:  
MSU Center for Orthopedic Research  
Ingham Regional Orthopedic Hospital  
2727 S. Pennsylvania Avenue,  
Lansing, MI 48910

Tel. 517-975-3302  
Fax 517-975-3305  
Email: cholewic@msu.edu